Author's response to reviews

Title: Severity of acute hepatitis and its outcome in patients with dengue fever in a tertiary care hospital Karachi, Pakistan (South Asia):

Authors:

Om Parkash (om.parkash@aku.edu)
Aysha Almas (aysha.almas@aku.edu)
SM Wasim Jafri (wasim.jafri@aku.edu)
Saeed Hamid (saeed.hamid@aku.edu)
Jaweed Akhtar (jaweed.akhter@aku.edu)
Hasnain alishah (hasnain.alishah@aku.edu)

Version: 8 Date: 1 May 2010

Author's response to reviews: see over
Covering Letter

To

Production Team

The BioMed Central Editorial Team

Subject: Revised Comments for

MS: 2523688802666021
Severity of acute hepatitis and its outcome in patients with dengue fever in a tertiary care hospital Karachi, Pakistan (South Asia):

Om Parkash, Aysha Almas, S.M Wasim Jafri, Saeed Hamid, Jaweed Akhtar, Hasnain Alishah

I am extremely thankful to BMC series for accepting my article in BMC gastroenterology.

1. Title: It has been changed as suggested
2. Tables: Tables have been added in main manuscript in table section and should not be used as additional files. And also Table has been amended as suggested formatting for tables.
3. Structure: We have renamed the introduction to background.
4. Abstract: Suggested headings have been removed from abstract section.
5. References: All citation have been placed before full stop.
6. References: lists of references have been updated as suggested and we have removed etal.
7. Figures: Image file does not contain anymore legend or figure number, however titles have been added after reference list in the section of figure legend.
8. Typography: has been checked.

Thanks

Regards

Dr Om Parkash